Author's response to reviews

Title: Barriers to successful recruitment of parents of overweight children for an obesity prevention intervention: a qualitative study among youth health care professionals

Authors:

Sanne MPL Gerards (sanne.gerards@maastrichtuniversity.nl)
Pieter C Dagnelie (dagnelie@maastrichtuniversity.nl)
Maria WJ Jansen (Maria.Jansen@ggdzl.nl)
Nanne K De Vries (n.devries@maastrichtuniversity.nl)
Stef PJ Kremers (s.kremers@maastrichtuniversity.nl)

Version: 2 Date: 13 October 2011

Author's response to reviews:

Dear editor,

We hereby send you a revised version of the paper entitled 'Barriers to successful recruitment of parents of overweight children for an obesity prevention intervention: a qualitative study among youth health care professionals'. We hope for the manuscript to be considered for publication in 'BMC Family Practice'.

The following revision was made. In the method section of the manuscript, on page 6, we included a sentence concerning the ethic statement: 'The interviews were part of a larger project which was approved by the ethical committee of the Maastricht University Medical Centre (trial number 10-3-052).'

Thanks in advance for your consideration,

On behalf of the coauthors,

Sanne Gerards